Author's response to reviews

Title: Truncated LEF-1 is one of the key regulators in the growth of colon carcinoma

Authors:

Shu-Hong Wang (wsh2003@126.com)
Ke-Jun Nan (nankj@163.com)
Yao-Chun Wang (Tanking@sina.com)
Wen-Juan Wang (Wangwenjuan1983@tom.com)
Tao Tian (Tiantao0607@163.com)

Version: 2 Date: 30 March 2011

Author's response to reviews: see over
Dear Editors:

We would like to submit our manuscript entitled “Truncated LEF-1 is one of the key regulators in the growth of colon carcinoma” to BMC Gastroenterology for your consideration of publication. The candidate target for a treatment of colon cancer has long been an unresolved and hot research issue. In this study, we demonstrated that the balance between the full and truncated forms of LEF-1 may play an important role in colon carcinogenesis. By using molecular analyses, we further showed that LEF-1 influence the colon cancer cell phenotype, including cell proliferation, colony formation and cell invasion. These results might be important in understanding the function and mechanism of LEF-1 in colon cancers, and provide a new direction for the therapy.

Thank you for your consideration.

Yours sincerely

Shu-Hong Wang

Department of Medical Oncology

The First Affiliated Hospital of The School of Medicine of Xi'an Jiaotong University

Xi'an 710061

China

Tel. +86-29-85324086

Email wsh2003@126.com